UNIVERSITY of York

This is a repository copy of *Promoting Positive Adolescent Mental Health (PPAMH!)* for School Climate:protocol for a feasibility randomised control trial in secondary schools [version 1; peer review: awaiting peer review].

White Rose Research Online URL for this paper: <u>https://eprints.whiterose.ac.uk/218892/</u>

Version: Accepted Version

Article:

Wadman, Ruth Elizabeth orcid.org/0000-0003-4205-0631, Taylor, Olivia orcid.org/0000-0001-7043-1795, Hudson, Kristian et al. (6 more authors) (2024) Promoting Positive Adolescent Mental Health (PPAMH!) for School Climate:protocol for a feasibility randomised control trial in secondary schools [version 1; peer review: awaiting peer review]. NIHR Open Research. 66. ISSN 2633-4402

https://doi.org/10.3310/nihropenres.13678.1

Reuse

This article is distributed under the terms of the Creative Commons Attribution (CC BY) licence. This licence allows you to distribute, remix, tweak, and build upon the work, even commercially, as long as you credit the authors for the original work. More information and the full terms of the licence here: https://creativecommons.org/licenses/

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



eprints@whiterose.ac.uk https://eprints.whiterose.ac.uk/

STUDY PROTOCOL

Promoting Positive Adolescent Mental Health (PPAMH!) for School Climate: protocol for a feasibility randomised control trial in secondary schools [version 1; peer review: awaiting

peer review]

Ruth Wadman¹, Olivia Taylor¹, Kristian Hudson², Dan Lewer², Catherine Hewitt¹, John Pickavance², Joseph Pryce³, John Wright², Simon Gilbody¹

¹Health Sciences, University of York, York, England, YO10 5DD, UK

²Bradford Teaching Hospitals NHS Foundation Trust, Bradford Institute for Health Research, Bradford, England, BD9 6RJ, UK
³London School of Hygiene & Tropical Medicine, London, England, WC1E 7HT, UK

V1 First published: 24 Oct 2024, 4:66 https://doi.org/10.3310/nihropenres.13678.1 Latest published: 24 Oct 2024, 4:66 https://doi.org/10.3310/nihropenres.13678.1

Open Peer Review

Approval Status AWAITING PEER REVIEW

Any reports and responses or comments on the article can be found at the end of the article.

Abstract

Introduction

We describe a feasibility randomised controlled trial to determine if it is possible to implement a full cluster-randomised trial of a multicomponent whole-school intervention aiming to improve the school climate. The Promoting Positive Adolescent Mental Health (PPAMH!) intervention will be adapted from an intervention previously trialled in India and Australia that reduced bullying, depression and risk behaviours among students.

Methods

We will randomise six secondary schools in Bradford in a feasibility trial. Three schools will receive the intervention and three will act as the 'usual care' control condition. We will collect quantitative and qualitative data on the implementation through activity logs and interviews with staff and well-being mentors. We will determine the feasibility of accessing and using routine education data and selfreport mental health data as outcomes in the full trial.

Results



This study will provide information about the acceptability and feasibility of delivering and evaluating the school climate intervention in secondary schools.

Conclusions

This is the first study to deliver a promising whole school climate intervention in the UK. The results of this feasibility trial will inform the decision to embark on a full trial and further intervention adaptations as needed.

Trial registration

ISRCTN registry reference ISRCTN14856201

Plain English Summary

PPAMH! ('Promoting Positive Adolescent Mental Health) for School Climate encourages a positive environment to support student wellbeing. A 'well-being mentor' will work in each school for half of the week. They will do whole-school activities such as assemblies, peer group activities such as workshops, and one-to-one activities with pupils. The intervention focuses on aiding pupils to have their say on school policies and curriculum by creating a participative school environment.

Interventions using a similar approach have been done in India and Australia. We think that PPAMH! could improve student attendance. The research trial in India found that it reduced bullying and feelings of depression. We will adapt the trial to be suitable for schools in Bradford through workshops with students and staff.

This initial trial will involve six secondary schools in Bradford. Schools will be randomly chosen to either being a control school or to receive the intervention. The main outcome is the successful running of the intervention in the schools, for example looking at how many activities were completed and by how many students. We will not study the effect of the intervention on school absences or other aspects of wellbeing. We will interview school staff members who are involved in providing the intervention in the schools and the well-being mentors. This will tell us about barriers to the intervention, how these could be overcome and changes that schools make to the intervention. We will also see how well we are able to access the data we plan to use in the full trial, which is school absences (from routine educational data) and mental health data such as depression and anxiety symptoms (via student surveys). We will plan to do a full trial if the feasibility trial shows the PPAMH! intervention can be successfully set up in schools.

Keywords

School based interventions, School attendance, Mental health, Secondary schools

Corresponding author: Ruth Wadman (ruth.wadman@york.ac.uk)

Author roles: Wadman R: Conceptualization, Funding Acquisition, Methodology, Project Administration, Resources, Supervision, Writing – Original Draft Preparation, Writing – Review & Editing; **Taylor O**: Conceptualization, Methodology, Project Administration, Resources, Writing – Original Draft Preparation, Writing – Review & Editing; **Hudson K**: Methodology, Project Administration, Resources, Writing – Original Draft Preparation, Writing – Review & Editing; **Lewer D**: Conceptualization, Funding Acquisition, Methodology, Supervision, Writing – Original Draft Preparation, Writing – Review & Editing; **Lewer D**: Conceptualization, Funding Acquisition, Methodology, Supervision, Writing – Original Draft Preparation, Writing – Review & Editing; **Hewitt C**: Methodology, Supervision, Writing – Review & Editing; **Pickavance J**: Methodology, Project Administration, Writing – Review & Editing; **Pryce J**: Conceptualization, Methodology, Writing – Review & Editing; **Wright J**: Conceptualization, Funding Acquisition, Resources, Supervision, Writing – Review & Editing; **Gilbody S**: Conceptualization, Funding Acquisition, Methodology, Supervision, Writing – Review & Editing; **Gilbody S**: Conceptualization, Funding Acquisition, Methodology, Supervision, Writing – Review & Editing; **Gilbody S**: Conceptualization, Funding Acquisition, Methodology, Supervision, Writing – Review & Editing

Competing interests: No competing interests were disclosed.

Grant information: This project is funded by the National Institute for Health Research (NIHR) under its ['Research for Patient Benefit (RfPB) Programme' (Grant Reference Number PB-PG- NIHR205448)]. The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care.

The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Copyright: © 2024 Wadman R *et al.* This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

How to cite this article: Wadman R, Taylor O, Hudson K *et al.* Promoting Positive Adolescent Mental Health (PPAMH!) for School Climate: protocol for a feasibility randomised control trial in secondary schools [version 1; peer review: awaiting peer review] NIHR Open Research 2024, **4**:66 https://doi.org/10.3310/nihropenres.13678.1

First published: 24 Oct 2024, 4:66 https://doi.org/10.3310/nihropenres.13678.1

The prevalence of mental health problems is increasing in young people. In the UK, rates of diagnosis for anxiety, depression, attention disorders, and eating disorders doubled between 2003 and 2018¹, while the COVID-19 pandemic and associated restrictions likely had an additional impact². The Mental Health of Children and Young People in England 2022 report found that 18% of 7- to 16-year-olds and 22% of 17- to 24-year-olds had a probable mental health disorder³. In addition to the impact of mental health problems on young people and their families, young people's mental health difficulties are associated with large economic costs and impacts on public services including the NHS⁴. Increasing demand for, and pressures on, specialist services such as Child and Adolescent Mental Health Services (CAMHS)⁵, has led to young people facing longer waiting times or having no contact with support services.

Most mental health problems start by adolescence, with half established by age 14⁶⁻⁷. Elevated mental health symptoms and diagnoses among adolescents are predictive of later mental health problems and are associated with worse outcomes in multiple areas of daily life in adulthood, including educational achievements and employment patterns^{8–11}. Children and young people with mental health problems are at risk of higher school absence rates. In England in 2021, 13% of children with a probable mental disorder missed more than 15 days of school in one year, compared with 4% of those without a mental health disorder¹².

Schools play a crucial role in supporting wellbeing and mental health, with children spending more time in school than any other formal institution. School-based preventative interventions for mental health include universal approaches (for all students), selective interventions targeted at at-risk groups, and indicative (early) interventions for individuals with mental health symptoms.

Universal social and emotional learning (SEL) interventions, that aim to improve social and emotional knowledge and skills, have been found to reduce symptoms of depression and anxiety at least in the short-term but there is limited evidence of the longer-term effect of SEL¹³. Evidence also indicates that targeted depression- and anxiety-focused school-based interventions, such as targeted cognitive behavioural therapy (CBT) delivered by external experts, can reduce symptoms in the short- and medium-term^{14,15}. However, the effectiveness of universal school-based CBT preventative interventions is less apparent, with two trials reporting no improvements in mental health symptoms^{16,17}.

These large-scale evaluations present a mixed picture for the effectiveness of universal school-based interventions. A UK trial of classroom-based mindfulness (MYRIAD trial) found no evidence of benefit in terms of reducing risk of depression or improving socioemotional-behavioural functioning¹⁸ and additional analyses suggested the intervention may harm students at risk of mental health problems¹⁹. Findings like these have led to calls for researchers to recognise that universal school-based interventions can lead to negative

outcomes in children and young people and to look for different intervention $models^{20}$.

There are some multicomponent whole-school preventative mental health interventions based on the World Health Organisation 'Health Promoting Schools' framework. A recent cluster-RCT evaluated a multicomponent whole-school health promotion intervention delivered in secondary schools by lay counsellors in Bihar, India (SEHER trial). They reported significant improvements in comparison to a control group in measures of school climate, depression, attitudes towards gender equity, bullying and victimisation^{21,22}. Significant effects on these outcomes were observed at the end of year 1, with further improvements observed after two years. The study further reported that students in intervention schools enjoyed the activities and found them interesting and informative. Previously, an RCT of a similar intervention in Australia (the Gatehouse project) observed a significant reduction in adolescent risk behaviours such as drinking and regular smoking^{23,24}. We aim to explore the applicability of this 'Health Promoting Schools' framework in the UK.

We will work with schools and young people to adapt the SEHER intervention and intervention resources to be appropriate and acceptable to secondary schools in a city in the UK. We will test the adapted intervention and its implementation in a feasibility trial before doing a full trial (aiming to include all secondary schools). A key factor is whether a full trial can be done using mental health data collected as part of the Born in Bradford cohort study (further details to follow).

Aims and objectives

The study aim is to carry out a feasibility trial to determine if it is possible to implement a cluster-randomised trial of a multi-component whole-school intervention aiming to improve the school climate (PPAMH!). The feasibility trial will not aim to estimate the effect of the intervention, but to learn whether the intervention and research are feasible. A full trial is planned if the intervention is feasible, and the resources are identified.

The specific objectives are to:

1. Establish whether secondary schools can be recruited to take part in the trial.

2. Pilot the randomisation process, including accessing the school-level data for the factors used in randomisation.

3. Assess the acceptability of the PPAMH! intervention to school staff and the well-being mentors involved in implementation using qualitative research methods, including barriers faced and possible solutions.

4. Assess adherence to the PPAMH! intervention by collecting data on the activities undertaken in school (e.g. number of sessions and number attending).

5. Establish whether school absence data and mental health data can be accessed via the education record and the Born in Bradford cohort study respectively.

The design is a feasibility cluster-randomised trial of an intervention that aims to create a more positive school environment. The feasibility trial will involve randomisation of six schools with a primary outcome of the successful delivery of the intervention in schools determined by 1) qualitative interview data from staff involved in implementing the intervention and 2) quantitative data on intervention delivery via an activity log including number of sessions and attendees.

The quantitative outcome data we will access in this study is either routine data from the education record or is data that is already being collected as part of the Born in Bradford cohort study. The cohort study has HRA ethical approval (IRAS number 295413; REC reference number 21/YH/0261) and this approval includes plans for supporting interventions that will use Born in Bradford outcome data. The Born in Bradford study information and informed consent process included information about the use of participant data to evaluate interventions, to ensure that cohort participants are fully informed.

Research ethics

The study has been reviewed by the Health Sciences' Research Governance Committee (University of York) and received approval on 15th March 2024 (Ref: HSRGC/2024/616/F). Informed written consent from participating school leaders, school staff and wellbeing mentors will be obtained by the research team.

Study outcomes

Primary outcome

Successful delivery of the intervention in schools determined by descriptive quantitative data on intervention delivery via an activity log of number of sessions, frequency of sessions, number of attendees and evidence of training to support delivery.

Assessment of qualitative data about implementation via interviews with relevant school staff (and the wellbeing mentors) involved in implementing the intervention.

Secondary outcomes

Successful collection of school absences data, as a proxy measure of the school environment. This was selected as the planned primary outcome for the full trial by young people and stakeholders. It will be available for all pupils via the educational record (routine data).

Access to mental health and wellbeing data of year 9 students, collected in the Born in Bradford cohort study, including mental health problems such as depression and anxiety (measured by RCADS-25), mental wellbeing (measured by SWEMWBS) and bullying.

Methods

Study setting

The study is located in Bradford, an urban, multicultural city in the North of England, UK. Bradford is one of the largest local authorities in England with a population of 547,000, and has one of the youngest and most diverse populations in the UK²⁵. Within Bradford there are 59 secondary schools, including 7 maintained schools, 31 academies, 11 independent schools, 9 special schools and 1 independent sixth form college.

This study will make use of survey mental health data being collected in Bradford secondary schools as part of the Born in Bradford cohort study²⁶. Born in Bradford's 'Age of Wonder' study is a continuation and expansion of the Born in Bradford longitudinal birth cohort study that began in 2007. Between 2007 and 2011, 12,453 women were recruited who delivered 13,786 live infants²⁷. Half of the Born in Bradford participants live within the fifth most deprived wards for England and Wales, and 45% are of Pakistani origin²⁸. From 2021, Born in Bradford have expanded the cohort to include up to 30,000 young people, with the aim of following these young people through adolescence and into adulthood.

Preliminary results from the Born in Bradford survey of young people in 2022–23 (N \sim 5,000) suggest the prevalence of depression and anxiety in young people is similar to other parts of England, and rates of self-harm and probable eating disorders elevated²⁹. Additionally, school absences and persistent absences (defined as missing 10% or more of school) are higher than regional and national averages, with absence rates continuing to rise despite improvements being seen in other parts of Yorkshire and England³⁰.

Patient and public involvement

A group of young people (aged 16 to 24), The Bradford Healthy Minds Apprentices were consulted at the design phase of the study in a workshop held in September 2022. This group met again in February 2023 to decide upon the name of the intervention and to design the study logo. In May 2023, the group met to review the implementation details for the intervention including who should deliver the intervention in schools. There will be additional input from young people and school staff to adapt the intervention (details below).

Eligibility criteria

Inclusion criteria Secondary school in Bradford.

Actively participating in the Born in Bradford cohort study.

Exclusion criteria

School is outside of Bradford.

School is not part of the Born in Bradford cohort study.

Recruitment

Schools will be recruited via existing research links with secondaries in Bradford through the Born in Bradford cohort study²⁶. School staff participants will be recruited once the intervention is underway and the staff relevant to the intervention implementation are identified. Intervention delivery will take place in the schools and will be led by 'wellbeing mentors': non-clinical community workers.

PPAMH! intervention

A school climate intervention was identified as a priority by young people and stakeholders in Bradford. The intervention will be adapted for each school through research involvement activities and workshops with young people and key stakeholders such as school staff. For example, students will be asked to choose the topics of focus in the intervention for each half-term (i.e. six weeks).

The PPAMH! intervention is a multi-level whole school intervention. The intervention's conceptual framework emphasises the importance of a positive school climate. The name of this adapted intervention 'Promoting Positive Adolescent Mental Health (PPAMH!) for School Climate' was chosen by our young person's advisory group.

A positive school climate can be described as one that features a sense of belonging to the school, a participative school environment, student commitment to academic values, and supportive relationships among school students, teachers, parents, and leadership. Improvements to the school climate are targeted using a multi-level school-based intervention, organised at the whole-school, peer group and individual levels.

PPAMH! includes four priority areas: promoting social skills; engaging the school community (i.e., pupils, teachers, and parents) in school decision-making processes; improving understanding of health and risk behaviours; and enhancing problem-solving skills. By promoting a positive school climate, supportive relationships and enhancing emotional and mental health knowledge, the intervention aims to improve student mental health and wellbeing (i.e. reduced feelings of depression and less bullying)²¹.

The PPAMH! intervention itself will be delivered by two voluntary sector non-clinical community workers ('well-being

mentors'), employed by a local mental health charity and supported by the local NHS Trust. In addition to oversight from the charity, the research team will provide training and regular meetings and site visits for the wellbeing mentors.

A well-being mentor will be in intervention schools 50% of the school week to facilitate whole school activities. They will also run peer group sessions and 1-1 activities with year 9 pupils. Example activities are awareness generation, policy review, workshops and skills training (see Table 1). Each halfterm (six weeks) will have a different topic of focus, which will have been selected by the students and school staff. Example topics may include mental health and wellbeing, study skills, and rights and responsibilities.

Adapting the intervention through PPIE

The intervention needs to be adapted for Bradford, including the intervention components (examples are given in Table 1) and the support that will be provided to schools. We will work with young people and schools to develop an intervention handbook and training package. For every intervention school, we aim to hold three one-hour workshops with secondary school students (10-15 participants per workshop) and three one-hour workshops with relevant school staff (6-8 participants per workshop). These will be in-person at school or online, depending on the preferences of the attendees. Workshop 1 will prioritize the topics to be covered as part of the intervention, such as mental health, relationships and rights and responsibilities. Workshop 2 will seek feedback on the proposed activities that form the intervention. Workshop 3 will discuss the delivery of the intervention and how to engage students, such as assemblies, topic boards and workshops. This information will be used to adapt the intervention handbook and resources. The number of workshops held will depend on the timing of recruitment and randomisation of schools. A pragmatic approach will be

Whole School Level	Peer Group Level	Individual Level
Formation of a School Health Promotion Committee	Elected group meetings to discuss peer concerns, develop plans of action, and organise activities	One-to-one support with problems
Assembly-based skits/ performances	Workshops (eg. study skills for pupils, teacher discipline practices)	Referral to appropriate services
Monthly wall magazine to build knowledge on key issues		
Speak-out box (concerns raised and addressed)		

Table 1. Examples of specific activities undertaken to improve school climate in previous trials.

taken and fewer workshops may be necessary to fit with the school timetable.

Randomisation

Schools will be randomised to either being a control school (three schools) or receiving the intervention (three schools). Schools randomised to the control arm of the pilot trial will not be required to do anything additional. The control condition is 'usual care'.

Schools will be randomised using minimisation. Minimisation will be undertaken in minimPy (or equivalent package in R) using naïve minimisation. Minimisation factors will include 1) baseline absence rate, 2) number of pupils from non-White ethnic groups and 3) number of female pupils.

The proposed timeline for the study is given in Figure 1. Staff and student workshops with the three intervention schools, where we will adapt the intervention, will take place in the Autumn term 2024 (September through to December).

Feasibility of outcome data collection

Should a full trial take place, we will describe characteristics and baseline outcome measures for schools and pupils using routinely collected data (which is generally publicly available via systems such as Get Information about Schools - GOV.UK get-information-schools.service.gov.uk). The primary outcome for the full trial will be school absences taken from the education record (routine data). This will be provided at term and pupil level by participating schools, with minimal additional pupil level variables. We will also collect data on age, year group, sex, ethnicity and eligibility for free school meals. Secondary outcomes include depression and anxiety measured using the Revised Child Anxiety and Depression Scale (RCADS-25)³¹, wellbeing measured using the Short Warwick-Edinburgh Mental Wellbeing Scale (SWEMWEBS)³² and bullying (bespoke questions). These data are being collected in Bradford schools as part of the Born in Bradford cohort study. For the feasibility trial, we will just determine how feasible it is to access and use this data (i.e. we will not make any comparison in outcomes between intervention and control schools).

Implementation data collection

Relevant members of school staff who are involved in the intervention implementation will be invited to take part in interviews: one initial long form interview and a series of rapid 15-minute interviews (repeated approximately every six weeks). We would ideally include all staff members involved in implementing the intervention, but at a minimum we would recruit three members of school staff in each school. The 'wellbeing mentors' will also be interviewed and in addition will be asked to keep a log of PPAMH! activities undertaken in the school as part of the implementation evaluation. This will include data on the number of sessions, frequency of sessions, number of attendees, evidence of training to support delivery, any adaptations made by the school to the intervention or its delivery, any implementation strategies they have used, and any other relevant information identified by our implementation specialist.

Analysis

Quantitative analysis

The feasibility trial will not include formal statistical analysis. We will describe characteristics of participating schools and the proportion of missing data for the outcomes. We will summarise the quantitative implementation data descriptively. If the results of the feasibility trial show that no major changes are required to the intervention, a full trial is planned. We will publish a protocol for the full trial with detailed analysis plans.

Qualitative analysis

The initial long-form interview will be transcribed. The rapid response interviews will be audio recorded, listened to, and then summarised into lightning reports. The lightning reports will summarise what is going well, what is not going so well, and any key insights participants have in regard to PPAMH! and its implementation. Lightning reports and long-form interview data will then be coded using the Consolidated Framework for Implementation Research (CFIR) 2.0 and The Expert Recommendations for Implementing Change (ERIC) list. The CFIR analysis will result in a list of barriers and facilitators to implementation. The ERIC will highlight which implementation strategies sites have used to implement PPAMH! and overcome contextual barriers. Adoption and acceptability will also be captured the interview data by asking participants about these directly. Sustainability will be measured in the final rapid interview using the Program Sustainability Assessment Tool, a 15 minute sustainability questionnaire.

Determinants within the CFIR, the implementation strategies listed in ERIC and adoption, and acceptability will be used as codes in NVivo to produce a framework matrix that summarises the information housed under each code at each site. CFIR valence and strength coding will allow us to prioritise barriers



Figure 1. PPAMH! proposed timeline.

and facilitators. We will identify the implementation strategies used to address these. We will then aim to provide a rationale (causal mechanism) for how these strategies worked to address barriers and contribute to our collected implementation outcomes and present this data within implementation research logic models (IRLMs) for each site.

Progression to a full trial

Data collected on the activities undertaken in schools as part of the intervention, along with the qualitative implementation data, will be used to determine whether a full trial will take place. Each school will have a target for the number of intervention activities completed each half-term/year adapted from the original SEHER trial, for example four awareness-generating assembles per month and two health promotion committee meetings per year²¹. The intervention activity logs will be compared against these targets. The decision to progress to a full trial will be based on whether these targets have largely been met and if the qualitative analysis of implementation data strongly suggests that school staff and well-being mentors thought the intervention is acceptable and can be implemented in school.

If the quantitative outcome data can be effectively accessed as part of the feasibility trial, then this outcome data will be included in the full trial (i.e. as an internal pilot). If the data access is not feasible or limited due to missing data, then the schools included in the feasibility trial will not be included in the full trial (i.e. an external pilot)³³. This will also be the case if the implementation analysis suggests that a substantial amount of modification to the PPAMH! intervention is required for it to be successfully implemented in schools.

Adverse event reporting

Possible harm as a result of the study will be monitored (e.g. mental distress including self-harm/suicidal behaviour). Standard operating procedures for reporting adverse events will be adapted by the study team. The wellbeing mentors will be prompted to record any adverse events related to the intervention as part of their intervention activity log. The schools' existing safe-guarding policies will be adhered to. The Trial Steering Committee will review adverse events throughout the trial for safety. If there is evidence of harm due to the intervention or measures used, this may result in a possible recommendation to stop the trial.

Study limitations

Due to the nature of the intervention it is not possible for schools to be blinded to study allocation. If a full trial is established, the analyst/statistician will be blinded to the intervention status of each school.

Secondary outcomes from the Born in Bradford cohort study are likely to have missing data, principally because of organisational issues in the participating schools on the days of data collection and also due to some pupils being absent on the day of the survey.

Discussion

This study aims to examine the feasibility of implementing a multi-component whole school intervention for improving the school climate in Bradford secondary schools. The proposed outcome measures of school absence and mental health will be taken from routine data and the Born in Bradford cohort study. This study will explore the feasibility of implementing the intervention and accessing the required data for a full trial. The PPAMH! intervention will be adapted from the SEHER programme trialled in Bihar, India. SEHER has shown promising results in improving the school climate and reducing symptoms of depression^{21–22}. This is the first time this intervention has been implemented in the UK. The results will inform the decision to move to a full trial and will add to the body of literature evaluating school-based mental health interventions.

Ethics and consent

The study has been reviewed by the Health Sciences' Research Governance Committee (University of York) and received approval on 15th March 2024 (Ref: HSRGC/2024/616/F). Informed written consent from participating school leaders, school staff and wellbeing mentors will be obtained by the research team.

Data availability

Underlying data No data are associated with this article.

Extended data

Figshare: Repository: PPAMH! for School Climate feasibility trial measures, "https://doi.org/10.6084/m9.figshare.27233055

The project contains the following files:

• The survey questions

Figshare: Study materials PPAMH! for school climate feasibility protocol, https://figshare.com/articles/online_resource/ Study_materials_PPAMH_for_school_climate_feasibility_ protocol/27188265?file=49670706

The project contains the following files:

- Information sheets
- Consent form

Reporting guidelines

Figshare: SPIRIT Checklist for PPAMH! for School Climate feasibility protocol, https://doi.org/10.6084/m9.figshare.26325616

The project contains the following files:

• Spirit checklist

Data are available under the terms of the Creative Commons Attribution 4.0 International license (CC-BY 4.0) (https://creativecommons.org/licenses/by/4.0/).

The Program Sustainability Assessment Tool is available online: https://www.sustaintool.org/psat/assess/

Acknowledgements

This report is independent research funded by the National Institute for Health and Care Research Yorkshire and Humber Applied Research Collaboration. The views expressed in this publication are those of the author(s) and not necessarily those of the National Institute for Health and Care Research or the Department of Health and Social Care.

We are very grateful to the Bradford Healthy Minds Apprentices who were involved in the Patient and Public Involvement activities for this study.

References

- Cybulski L, Ashcroft DM, Carr MJ, et al.: Temporal trends in annual incidence rates for psychiatric disorders and self-harm among children and adolescents in the uk, 2003–2018. BMC Psychiatry. 2021; 21(1): 229. PubMed Abstract | Publisher Full Text | Free Full Text
- Hafstad GS, Augusti EM: A lost generation? COVID-19 and adolescent mental health. Lancet Psychiatry. 2021; 8(8): 640–1.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Newlove-Delgado T, Marcheselli F, Williams T, et al.: Mental health of children and young people in England. NHS Digital, Leeds. 2022. Reference Source
- Annual Report of the Chief Medical Officer 2012: Our children deserve better: prevention pays. 2012; [March 2024]. Reference Source
- 5. Children's Commissioner: Children's mental health services 2021–22. 2023 [March 2023].
 - Reference Source
- Kessler RC, Berglund P, Demler O, et al.: Lifetime prevalence and age-ofonset distributions of DSM-IV disorders in the national comorbidity survey replication. Arch Gen psychiatry. 2005; 62(6): 593–602. PubMed Abstract | Publisher Full Text
- Solmi M, Radua J, Olivola M, et al.: Age at onset of mental disorders worldwide: large-scale meta-analysis of 192 epidemiological studies. Mol Psychiatry. 2022; 27(1): 281–95.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Patton GC, Coffey C, Romaniuk H, et al.: The prognosis of common mental disorders in adolescents: a 14-year prospective cohort study. Lancet. 2014; 383(9926): 1404–11.
 PubMed Abstract | Publisher Full Text
- Mulraney M, Coghill D, Bishop C, et al.: A systematic review of the persistence of childhood mental health problems into adulthood. Neurosci Biobehav Rev. 2021; 129: 182-205.
 PubMed Abstract | Publisher Full Text
- Chen H, Cohen P, Kasen S, et al.: Impact of adolescent mental disorders and physical illnesses on quality of life 17 years later. Arch PediatrAdolesc Med. 2006; 160(1): 93–9.
 PubMed Abstract | Publisher Full Text
- Pollard J, Reardon T, Williams C, et al.: The multifaceted consequences and economic costs of child anxiety problems: a systematic review and meta-analysis. JCPP Adv. 2023; 3(3): e12149.
 PubMed Abstract | Publisher Full Text | Free Full Text
- NHS Digital: "Mental health of children and young people in England, 2021: wave 2 follow up to the 2017 survey," NHS England, 2021. 2021; [March 2023].

Reference Source

- Van de Sande MCE, Fekkes M, Kocken PL, et al.: Do Universal Social and Emotional Learning programs for secondary school students enhance the competencies they address? A systematic review. Psychol Sch. 2019; 56(10): 1545–67.
 Publisher Full Text
- Clarke A, Sorgenfrei M, Mulcahy J, et al.: Adolescent mental health. A systematic review on the effectiveness of school-based intervention. London: Early Intervention Foundation. July, 2021. Reference Source
- Zhang Q, Wang J, Neitzel A: School-based mental health interventions targeting depression or anxiety: a meta-analysis of rigorous randomized controlled trials for school-aged children and adolescents. J Youth Adolesc. 2023; 52(1): 195–217.
 PubMed Abstract | Publisher Full Text | Free Full Text

- Andrews JL, Birrell L, Chapman C, et al.: Evaluating the effectiveness of a universal eHealth school-based prevention programme for depression and anxiety, and the moderating role of friendship network characteristics. *Psychol Med.* 2023; 53(11): 5042–51.
 PubMed Abstract | Publisher Full Text
- Stallard P, Phillips R, Montgomery AA, et al.: A cluster randomised controlled trial to determine the clinical effectiveness and cost-effectiveness of classroom-based Cognitive-Behavioural Therapy (CBT) in reducing symptoms of depression in high-risk adolescents. *Health Technol Assess*. 2013; **17**(47): vii-xvii, 1–109.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Kuyken W, Ball S, Crane C, et al.: Effectiveness and cost-effectiveness of Universal School-Based Mindfulness Training compared with normal school provision in reducing risk of mental health problems and promoting well-being in adolescence: the MYRIAD cluster randomised controlled trial. *BMJ Ment Health*. 2022; 25(3): 99–109.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Montero-Marin J, Allwood M, Ball S, et al.: School-Based Mindfulness Training in early adolescence: what works, for whom and how in the MYRIAD trial? Evid Based Ment Health. 2022; 25(3): 117–24.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Foulkes L, Andrews JL, Reardon T, et al.: Research recommendations for assessing potential harm from universal school-based mental health interventions. Nat Ment Health. 2024; 2: 270–7. Publisher Full Text
- Shinde S, Weiss HA, Varghese B, et al.: Promoting school climate and health outcomes with the SEHER multi-component secondary school intervention in Bihar, India: a cluster-randomised controlled trial. Lancet. 2018; 392(10163): 2465–77.
 PubMed Abstract | Publisher Full Text
- Shinde S, Weiss HA, Khandeparkar P, et al.: A multicomponent secondary school health promotion intervention and adolescent health: an extension of the SEHER cluster randomised controlled trial in Bihar, India. PLoS Med. 2020; 17(2): e1003021.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Bond L, Thomas L, Coffey C, et al.: Long-term impact of the gatehouse project on cannabis use of 16-year-olds in Australia. J Sch Health. 2004; 74(1): 23-9.
 PubMed Abstract | Publisher Full Text
- Bond L, Patton G, Glover S, et al.: The gatehouse project: can a multilevel school intervention affect emotional wellbeing and health risk behaviours? / Epidemiol Community Health. 2004; 58(12): 997–1003.
- behaviours? J Epidemiol Community Health. 2004; 58(12): 997–1003. PubMed Abstract | Publisher Full Text | Free Full Text 25. City of Bradford Metropolitan District Council: Bradford district population
- City of Bradford Metropolitan District Council: Bradford district population update. 2022 [March 2024]. Reference Source
- Shire KA, Newsham A, Rahman A, et al.: Born in Bradford's Age of Wonder cohort: protocol for adolescent data collection [version 1; peer review: 2 approved, 1 approved with reservations]. Wellcome Open Res. 2024; 9: 32. PubMed Abstract | Publisher Full Text | Free Full Text
- Raynor P, Born in Bradford Collaborative Group: Born in Bradford, a cohort study of babies born in Bradford, and their parents: protocol for the recruitment phase. BMC Public Health. 2008; 8: 327.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Wright J, Small N, Raynor P, et al.: Cohort profile: the Born in Bradford multi-ethnic family cohort study. Int J Epidemiol. 2013; 42(4): 978–91. PubMed Abstract | Publisher Full Text
- 29. Wadman R, Taylor O, Qualter P, et al.: A country that works for all children

and young people: improving mental health and wellbeing with and through education settings. April, 2024. Publisher Full Text

- 30. Create your own tables. Explore Education Statistics GOV.UK. (n.d.). Reference Source
- Ebesutani C, Reise SP, Chorpita BF, et al.: The Revised Child Anxiety and Depression Scale-short version: scale reduction via exploratory bifactor modeling of the broad anxiety factor. *Psychol Assess.* 2012; 24(4): 833–45. PubMed Abstract | Publisher Full Text
- 32. Stewart-Brown S, Tennant A, Tennant R, *et al.*: **Internal construct validity** of the Warwick-Edinburgh Mental Well-Being Scale (WEMWBS): a Rasch analysis using data from the Scottish Health Education Population Survey. *Health Qual Life Outcomes.* 2009; 7: 15. PubMed Abstract | Publisher Full Text | Free Full Text
- Avery KNL, Williamson PR, Gamble C, et al.: Informing efficient Randomised Controlled Trials: exploration of challenges in developing progression criteria for internal pilot studies. BMJ Open. 2017; 7(2): e013537. PubMed Abstract | Publisher Full Text | Free Full Text